Author's response to reviews

Title: Multi-Drug Resistant Gram Negative Infections and use of intravenous Polymyxin B in Critically ill Children of developing country: Retrospective cohort study

Authors:

Naveed-ur-Rehman Siddiqui Dr (naveed.rehman@aku.edu)
Farah Naz Qamar Dr (farah.qamar@aku.edu)
Humaira Jurair Dr (humaira.jurair@aku.edu)
Anwarul Haque Dr (anwar.haq@aku.edu)

Version: 4 Date: 7 October 2014

Author's response to reviews:

covering letter uploaded with reviewer comments